Author's response to reviews

Title: Antilock Breaking System Effectiveness in Prevention of Road Traffic Crashes in Iran

Authors:

Soheil Saadat Dr (soheil.saadat@gmail.com)
Davoud Khorasani Zavareh Dr (davoud.khorasani@gmail.com)
Saeed Shoar Dr (ssht84@yahoo.com)

Version: 2 Date: 12 July 2012

Author's response to reviews:

We did revise our manuscript based on the precious comments you provided point by point.
We hope this is just in the fair format you required.